Author's response to reviews

Title: Toward integrated and sustainable prevention against diabetes in rural China: study rationale and protocol of eCROPS

Authors:

Rui Feng Dr (xia0816meng@sina.cn)
Kaichun Li Dr (Lkc@lacdc.com.cn)
Jing Chai Dr (april0320chai@yahoo.com.cn)
Shaoyu Xie Dr (853571123@qq.com)
Jing Cheng Dr (Xuchi1982@yahoo.com.cn)
Pingfu Wei Dr (824169556@qq.com)
Debin Wang Dr (dbwang@vip.sina.com)

Version: 5 Date: 5 July 2013

Author's response to reviews:

We changed the section title of "significances" in the previous abstract into "discussion".